Author's response to reviews

Title: Study design of DIACORE (DIAbetes COhoRtE) - a cohort study of patients with diabetes mellitus type 2

Authors:

Lena Dörhöfer (lena.doerhoefer@ukr.de)
Alexander Lammert (Alexander.Lammert@umm.de)
Vera Krane (krane_v@medizin.uni-wuerzburg.de)
Mathias Gorski (mathias.gorski@ukr.de)
Bernhard Banas (bernhard.banas@ukr.de)
Christoph Wanner (wanner_c@medizin.uni-wuerzburg.de)
Bernhard K. Krämer (Bernhard.Kraemer@umm.de)
Iris M. Heid (iris.heid@ukr.de)
Carsten A. Böger (carsten.boeger@ukr.de)

Version: 2 Date: 11 December 2012

Author's response to reviews: see over
Dear Dr. Sands,

On behalf of our co-authors from the DIACORE Study Group, we are pleased to submit our revision of the manuscript (MS-ID: 4117866208157962) entitled

“Study design of DIACORE (DIAbetes COhoRtE) – a cohort study of patients with diabetes mellitus type 2 (DM2)” as “Study Protocol”.

As requested, we now provide a funding statement for the performance of the study in all centers (Universities of Regensburg und Mannheim). We also provide the requested ethics statement in the English language from both Universities.

At the time of submission of this revision, we have no other manuscript based on this protocol under review at any other journal.

This manuscript or parts of it are not under review at any other journal.

We look forward to hear from you. Yours sincerely,

Lena Dörhöfer, MD
University Medical Center Regensburg, Germany.
lena.doerhoefer@ukr.de

Carsten A. Böger, MD
University Medical Center Regensburg, Germany.
carsten.boeger@ukr.de